Massachusetts Institute of Technology Handout ?? 6.042J/18.062J: Mathematics for Computer Science May 9, 1994 Professors Charles E. Leiserson and Tom Leighton

Final Exam Information

When: Tuesday, May 17, 1994, 1:30{4:00 p.m.

Where: Athletic Center.

Crib sheets: The quiz is closed book. You may, however, bring two handwritten 8^1_2^{00} � 11^{00} sheets of paper as crib sheets. Preparing a crib sheet can be a useful study aid, so take time in selecting material for it. You may use both sides of the paper and write as small as you like, but in any case, you are allowed only two sheets.

Grades: Grades will be available from the course secretary, Cheryl Patton, in NE43{201, after Wednesday.